Author’s response to reviews

Title: Upregulation of MicroRNA-19b Predicts Good Prognosis in Patients with Hepatocellular Carcinoma Presenting with Vascular Invasion or Multifocal Disease

Authors:

Chung-Lin Hung (samhong1111@gmail.com)
Hung-Wen Tsai (hungwen@mail.ncku.edu.tw)
Chia-Jui Yen (yencj@mail.ncku.edu.tw)
Chia-Shen Yen (gsysycs@gmail.com)
Yu-Chieh Su (hepatoma@gmail.com)

Version: 4
Date: 26 June 2015

Author’s response to reviews: see over
Dear Dr. Magdalena Chechlinska, academic editors of BMC Cancer and reviewers:

This is a revision and resubmission of our manuscript. We have made some major modifications in response to the editor's request.

We are grateful for the opportunity to revise our work MS: 5180553861641647 entitled "Upregulation of MicroRNA-19b Predicts Good Prognosis in Patients with Hepatocellular Carcinoma Presenting with Vascular Invasion or Multifocal Diseases" and thank the academic editor and reviewers for the time and effort they took to critique our work and offer suggestions for improving it. We hope that you now find our manuscript suitable for publication in the BMC Cancer.

We look forward to your reply.

Sincerely yours,

Chia-Jui Yen, MD PhD
Corresponding Author

Division of Hematology and Oncology, Department of Internal Medicine
National Cheng Kung University Hospital
College of Medicine, National Cheng Kung University, Taiwan
Mailing address: 138, Sheng-Li Road, Tainan 704, Taiwan
E-mail: yencj@mail.ncku.edu.tw
EDITORIAL REQUEST:

Copyediting: We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service.

Reply: Thank you for the comments. We had copyedited our manuscript.